Author’s response to reviews

Title: Exercise training during pregnancy reduces circulating insulin levels in overweight/obese women postpartum. Secondary analysis of a randomised controlled trial (the ETIP trial).

Authors:

Kirsti Garnæs (kirsti.garnas@ntnu.no)
Siv Mørkved (siv.morkved@ntnu.no)
Kjell Salvesen (pepe.salvesen@ntnu.no)
Øyvind Salvesen (oyvind.salvesen@ntnu.no)
Trine Moholdt (trine.moholdt@ntnu.no)

Version: 1 Date: 15 Aug 2017

Author’s response to reviews:

Please see attached letter with response to reviewers.